Author's response to reviews

Title: VATS intraoperative tattooing to facilitate solitary pulmonary nodule resection

Authors:

Lourens Willekes (ljw898@hotmail.com)
Cherif Boutros (cherifboutros@hotmail.com)
Michael A Goldfarb (mgoldfarb@sbhcs.com)

Version: 4 Date: 17 January 2008

Author's response to reviews: see over
All required changes done as follow:

- Formatting changes requested

Title page: Please include a title page for the first page of the manuscript. This should include the following: the manuscript title (in sentence case i.e. remove all unnecessary capitals), the full author names (on one line) and beneath that, associated institutional affiliations for all authors (denoted by superscript numbers). The corresponding author details should then be listed, and a list of all author email addresses following that. The corresponding author should be denoted with the use of a symbol in the author list e.g. *.

- Title page created according to the guidelines

Abstract: Please include the heading `Abstract` for the abstract section.

- Done

Case description: Please include the section heading case description after the introduction section.

- Section heading Case description included

Materials and Methods: Please rename this section to `Methods`.

- Materials and methods renamed to Methods

Please ensure the abstract is identical in both the manuscript and submission system.

- Abstract Identical

Please use sentence case for all headings, such as change Figures Legends to Figure legends (lower case l).

- Done

Competing interests: Please include a Competing interests section above the authors contributions. Please move the statement "The authors declare that they have no competing interests" into the competing interests section.

- Competing interests section created
Authors' contributions - For the Authors' contributions we suggest the following kind of format before the Acknowledgements (please use initials to refer to each author's contribution): For example, AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

- **Done**

Acknowledgements - Please include an ‘Acknowledgements’ section between the Authors' contributions and the references. The manuscript should include a statement about obtaining consent for publication from the patient or their relative in the acknowledgements section. For example: "Written informed consent was obtained from the patient for publication of this Case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."

- **Done**

Reference list - The reference list should contain all authors' names, if there are 30 or below, the term ‘et al.’ should not be used. If there are more than 30 names a minimum of 30 should be listed, and then et al can be used.

- **Changes done accordingly**

Additional files - Please remove the additional file from the manuscript in the submission system

- **Done**

Thank you.